Author’s response to reviews

Title: Nontuberculous Mycobacterium Infection Complicated with Haemophagocytic Syndrome: A Case Report and Literature Review

Authors:

Wen Shi (ysfn_2005@126.com)
Yang Jiao (peterpunch@163.com)

Version: 4 Date: 28 Apr 2019

Author’s response to reviews:

Dear editor:

Thanks for your comments and suggestions. Here are our responses:

Comment from editor: I think the figure showing haemophagocytosis on bone marrow aspiration is important and would recommend the authors to put it in the manuscript as Figure 2. In addition, please add some formal description of histological findings, rather than just mention "haemophagocytosis".

According to your suggestion, we chose a high-quality picture of the patient’s bone marrow aspiration specimen showing typical haemophagocytosis and added it as our Figure 2 (In the manuscript, it will be the figure 1 in sequence). We also added more description as the figure legend: Bone marrow aspirate specimen showed a haemophagocyte engulfing a polymorphonuclear leukocyte, a lymphocyte and erythrocytes (Wright-Giemsa stain, 1000x).

The English language within our manuscript (this revised version) has been polished by American Journal Experts as recommended by the editor and we submitted the certificate as supplementary material 4 for editors’ reference. We sincerely thank you again for your comments and your patience. We hope our adjustment can provide more information about our case.
Yours sincerely,

Yang Jiao, M.D., M.P.H.
The Department of General Internal Medicine,
Peking Union Medical College Hospital, Peking Union Medical College and Chinese Academy of Medical Sciences
No.1 Shuaifuyuan, Dongcheng District, Beijing, China, 100730.
Tel: 86-10-69155645; E-mail: peterpumch@163.com